Author's response to reviews

Title: Development of a validation algorithm for 'present on admission' flagging

Authors:

Terri J Jackson (t.jackson@uq.edu.au)
Jude L Michel (judith.michel@uq.edu.au)
Rosemary Roberts (rroberts@aapt.net.au)
Jennie Shepheard (Jennie.Shepheard@dhs.vic.gov.au)
Diana Cheng (D.Cheng@latrobe.edu.au)
Julie Rust (J.Rust@usyd.edu.au)
Catherine Perry (Catherine.Perry@dhs.vic.gov.au)

Version: 7 Date: 2 November 2009

Author's response to reviews:

New note in text to alert readers to the accompanying supplementary files:

'The algorithm (written in SAS™ coding, see Additional files 1 and 2) was then used to estimate the rate of invalid flagging in the 2005/06 Victorian Admitted Episodes Dataset (VAED) of 2,031,666 de-identified patient episodes from Victorian public and private hospitals.'